Author’s response to reviews

Title: Revisiting Racial Differences in ESRD due to ADPKD in the United States

Authors:

Erin Murphy (erin.l.murphy@yale.edu)

Feng Dai (feng.dai@yale.edu)

Katrina Blount (katrina.blount@yale.edu)

Madeline Droher (madeline.drohere@yale.edu)

Lauren Liberti (lauren.liberti@yale.edu)

Deidra Crews (dcrews1@jhmi.edu)

Neera Dahl (neera.dahl@yale.edu)

Version: 1 Date: 05 Nov 2018

Author’s response to reviews:

We have amended the sections as requested.

1) The Declarations section has been revised

2) All authors emails are included

3) A conclusions section is marked.